Author's response to reviews

Title: Plasma concentrations of 25-hydroxyvitamin D among Jordanians: Effect of biological and habitual factors on vitamin D status

Authors:

Eyad M Mallah (emallah@uop.edu.jo)
Mohammad F Hamad (mhamad@uop.edu.jo)
Mays A ElManaseer (frozen_rose83@hotmail.com)
Nidal A Qinna (nqinna@uop.edu.jo)
Nasir M Idkaidek (nidkaidek@uop.edu.jo)
Tawfiq A Arafat (tarafat@uop.edu.jo)
Khalid Z Matalka (kzm@uop.edu.jo)

Version: 2 Date: 31 May 2011

Author's response to reviews:

The following statement was revised in page 4 (the method section) to say "The study has been approved by the Faculty of Pharmacy and Medical Sciences and the Deanship of Scientific Research at Petra University and was performed in compliance with the Helsinki Declaration following the approval of an independent ethics committee at Al-Mowasah Hospital in Amman, Jordan."